Author's response to reviews

Title: Intervention mapping for the development of a strategy to implement the insurance medicine guidelines for depression

Authors:

Feico Zwerver (f.zwerver@vumc.nl)
Antonius JM Schellart (ton.schellart@vumc.nl)
Johannes R Anema (h.anema@vumc.nl)
Kathelijne C Rammeloo (kathelijne.rammeloo@uwv.nl)
Allard J van der Beek (a.vanderbeek@vumc.nl)

Version: 6 Date: 3 January 2011

Author's response to reviews:

Amsterdam January 3, 2011

The BMC Series Editorial Production Team

MS: 1988645188347826 – Intervention mapping for the development of a strategy to implement the insurance medicine guidelines for depression

Dear Editor,

Thank you for your e-mail dated December 24, 2010, and all the best wishes for 2011.

Unfortunately I wasn’t able to respond to your mail earlier, because of a short holiday at the end of last year.

Concerning the abovementioned manuscript, we addressed your comments

I met all the major revisions in the manuscript

1) The title is identical in the manuscript and the submission system..

2) I put the abstract in the manuscript file and matches exactly what is in the submission system.

3) I changed reference number 20 into number 4, in the Legenda of Figure 1, and I cited reference 4 in the main text at the right place. Furthermore, I corrected the subsequent reference numbers in the main text and in the reference list.

4) I improved the manuscript for typographical errors

We hope that you will find this revised manuscript acceptable for publication in your journal.
Sincerely, also on behalf of the co-authors,

Feico Zwerver